Author’s response to reviews

Title: Comparison of Visual Results and Higher-order Aberrations After Small Incision Lenticule Extraction (SMILE): High Myopia vs. Mild to Moderate Myopia

Authors:

hongying jin (hongyingj@163.com)
Ting Wan (64230519@qq.com)
Fang Wu (fangwu@126.com)
Ke Yao (xlren@zju.edu.cn)

Version: 2 Date: 08 Jun 2017

Author’s response to reviews:

Dear editor,

We have modified the paper according to the review’s suggestions. We use red fonts in the paper to highlight the modification.

Mercè Morral (Reviewer 2): Include the details of the sample size calculation in the manuscript, and mention that the H-group may not have sufficient statistical power (n=20) to detect differences. Also, I agree with reviewer 1 that being HOAs the focus of the paper, sample size should be calculated accordingly

Answer: We describe the details of the sample size calculation in the manuscript and we use red fonts to highlight the modifications.

Yours Sincerely

Yao Ke, MD
Professor and Chief

Eye Institute of Zhejiang University

Eye Center, Second Affiliated Hospital, College of Medicine, Zhejiang University